\documentclass[]{article}
%\def\GW{/locgolem/GW}
\def\GW{/GW}
\def\visibility{true}
\input{\GW/root/GW4reports/GWcommons}

%\usepackage{a4wide}
\usepackage[a4paper, total={180mm,267mm}, left=15mm, top=15mm,]{geometry}

\usepackage[
%backend=biber,
backend=bibtex,
%backend=biber,
style=ieee,
%style=reading,
sorting=ydnt,
defernumbers
]{biblatex}
\urlstyle{sf}

\addbibresource{\GW/root/bibtex/golem.bib}
\addbibresource{\GW/root/bibtex/others.bib}


\begin{document}
\title{Research into Runaway electron physics basis on the small tokamak GOLEM \\brief summary of the REs@tGOLEM current status \\for the CAAS report}
\maketitle
\def\vcite#1{\nocite{#1}}

%\nocite{*}



\vcite{CerovskyJINST22}
\vcite{NovotnyJINST20}
\vcite{TunklMT22}
\vcite{MalecIEEE21} 
\vcite{MalecMT23}
\vcite{KulkovJINST22}


\printbibliography[]




\end{document}



